Author's response to reviews

Title: Factors associated with reporting nursing errors in Iran: a qualitative study

Authors:

fatemeh hashemi (hashemif@sums.ac.ir)
Alireza Nikbakht Nasrabadi (nikbakht@tums.ac.ir)
Fariba Asghari (fasghari@tums.ac.ir)

Version: 3 Date: 2 August 2012

Author's response to reviews: see over
Dear editor in chief of BMC nursing,

I would like to inform you that the manuscript was revised based on both reviewers’ comments:

Reviewer 8534449937377063: The methodology was improved and the mentioned details were added. It is well defined how the themes /categories were determined. Regarding the reviewer comments, the whole methodology was revised and the questions were answered one by one. The changes are highlighted in yellow.

The whole writing was improved and grammatical and punctuation issues were revised. Ambiguous phrases were made clear, as well.

Reviewer 6742197387378079: 1. The whole writing was improved. 2. It was tried to omit the repeated sentences. 3. The gap in the current knowledge and methods used for data rigors was explained more in the last paragraph of the methodology section. 4. The discussion part was revised. 5. The fifth comment was covered.